Author's response to reviews

Title: Identification of breast cancer patients with a high risk of developing brain metastases: A single-institutional retrospective analysis.

Authors:

Volker Rudat (volker.rudat@gmail.com)
Hamdan El-Sweilmeen (HEI-Sweilmeen@saad.com.sa)
Iris Brune-Erber (IErber@saadmedical.com)
Alaa A Nour (ANour@saadmedical.com)
Nidal Almasri (NMasri@saadmedical.com)
Saleh Altuwaijri (saltuwajri@saadmedical.com)
Elias Fadel (EFadel@saadmedical.com)

Version: 5
Date: 10 April 2014

Author's response to reviews:

Dear Editor,

As requested we
1. included the email addresses for all authors on the title page.
2. renamed our 'Introduction' as 'Background'.
3. included an acknowledgement section:

"We acknowledge the fund and support of this work by the SAAD Research & Development Center, Saad Specialist Hospital, and King Abdulaziz City of Science and Technology (KACST) (grant TSCR: 33-837). The views expressed in this publication are those of the authors."